Author's response to reviews

Title: Analysis of cystic fibrosis gene mutations in children with Cystic Fibrosis and in 964 infertile couples, first report from the Region of Basilicata, Italy.

Authors:

Domenico Dell'Edera PhD (domenicodelledera68@gmail.com)
Michele Benedetto PhD (michelesi@hotmail.it)
Gemma Gadaleda Pr. (gemma.gadaleta@uniba.it)
Domenico Carone MD (d.carone@inwind.it)
Donatello Salvatore MD (Donatello.Salvatore@libero.it)
Antonella Angione PhD (antonella.angione@regione.basilicata.it)
Massimiliano Gallo PhD (massimiliano.gallo@supporto.regione.basilicata.it)
Michele Milo PhD (michelemilo@alice.it)
Maria Laura Pisaturo MD (pisaml@tiscali.it)
Giuseppe Di Pierro MD (g.dipierro@alice.it)
Eleonora Mazzone MD (eleonora.mazzone@gmail.com)
Annunziata Anna Epifania PhD (nunziaepifania69@libero.it)

Version: 2 Date: 6 April 2014

Author's response to reviews:

I and my co-authors, we reviewed the following points:

1. Please include all the co-authors' email addresses in the Title page.

2. As your research involves human specimens and data, please could you confirm if an ethical approval was obtained and if it has been included the Methods section of the manuscript. If not, please include the name of the body which gave approval, with a reference number where appropriate. Any experimental research on humans must be in compliance with the Helsinki Declaration.

3. Please include an Authors' Contributions section at the end of the manuscript, before the reference list. We suggest the following kind of format (please use initials to refer to each author’s contribution).

4. Please include a figure title and legend section after the reference list.